Annals of Internal Medicine

ORIGINAL RESEARCH

Comparison of Semaglutide or Dulaglutide Versus Empagliflozin for Risk for Death and Cardiovascular Outcomes Among Patients With Type 2 Diabetes

Two Target Trial Emulation Studies

Anum Saeed, MD; Suresh R. Mulukutla, MD; Floyd Thoma, BS; Lara Lemon, PhD; Agnes Koczo, MD; Steven Reis, MD; Oscar Marroquin, MD; and Kevin Kip, PhD

Background: Reduction of premature death and adverse cardiovascular outcomes is a key goal in type 2 diabetes management.

Objective: To compare mortality and cardiovascular event risks in patients treated with semaglutide versus empagliflozin and, secondarily, dulaglutide versus empagliflozin.

Design: Target trial emulation studies from observational data comparing semaglutide- or dulaglutidetreated patients with propensity score-matched patients treated with empagliflozin.

Setting: Health care system of 703 academic and community clinical practices.

Participants: Patients aged 45 years or older with type 2 diabetes treated from 1 January 2019 to 31 December 2024 with semaglutide, dulaglutide, or empagliflozin.

Intervention: Initial treatment with semaglutide, dulaglutide, or empagliflozin. At baseline, concomitant treatment with other diabetes medication was permitted, excluding other glucagon-like peptide-1 receptor agonists or sodium-glucose cotransporter-2 inhibitors.

Measurements: A composite of death, myocardial infarction (MI), or stroke was the primary outcome, and

C ardiovascular disease (CVD) is the leading cause of morbidity and mortality in the United States. Although CVD is a generalized public health concern, patients with diabetes mellitus are at nearly double the risk for major adverse cardiovascular events (MACE) (1). Patients with diabetes have a higher likelihood of earlier onset of MACE and worse outcomes than those without diabetes (2, 3). This not only drastically affects patients' quality of life but also has major consequences

See also:

Summary for Patients

Web-Only Supplement secondary composite outcomes included death or MI, MI or stroke, and individual cardiac events.

Results: Patients treated with semaglutide (n = 7899) versus empagliflozin (n = 7899) were followed for a median of 2.2 years; the respective rates of the composite of death, MI, or stroke were 3.7% versus 4.5% at 2 years and 5.9% versus 6.9% at 3 years. Corresponding incidence rates for the composite outcome were 20.99 versus 23.56 per 1000 personyears, with a hazard ratio (HR) of 0.89 (95% CI, 0.78 to 1.02). The HRs for the individual outcomes were 0.97 (CI, 0.81 to 1.15) for death, 0.85 (CI, 0.68 to 1.05) for MI, and 0.62 (CI, 0.43 to 0.89) for stroke. Risks for dulaglutide- and empagliflozin-treated patients were similar for the composite outcome (HR, 1.03 [CI, 0.90 to 1.16]) and for death, MI, and stroke separately.

Limitation: Observational study design, lack of data on cause-specific mortality, and residual confounding.

Conclusion: Semaglutide treatment seems to confer some advantage over empagliflozin. This advantage was not observed for dulaglutide.

Primary Funding Source: American Heart Association.

Ann Intern Med. doi:10.7326/ANNALS-24-00775 For author, article, and disclosure information, see end of text. This article was published at Annals.org on 17 June 2025.

for productivity and overall health care costs (1). There is an urgent need to optimize allocation and use of preventive treatments in those with diabetes (2).

The advent of newer diabetes management classes, including glucagon-like peptide-1 receptor (GLP-1) agonists, dipeptidyl peptidase-4 inhibitors, and sodium-glucose cotransporter-2 (SGLT-2) inhibitors (4-6), has ushered diabetes management and prevention of MACE into a new era. Both SGLT-2 inhibitors and GLP-1 agonists have shown cardiovascular benefits in patients with type 2 diabetes mellitus and atherosclerotic CVD regardless of hemoglobin A_{1c} (Hb A_{1c}) levels (7, 8).

Multisociety guidelines, including from the American Heart Association and American College of Cardiology, now recommend the use of both of these classes regularly for patients with high risk for CVD and diabetes regardless of HbA_{1c} levels. Recent pooled

data suggest that all-cause mortality and MACE are alleviated with SGLT-2 inhibitors and GLP-1 agonists, but not with dipeptidyl peptidase-4 inhibitors, compared with usual care (9). However, few studies have examined these mechanistically different classes, and specifically the most used drugs in each class, in a head-to-head analysis. Understanding the comparative effectiveness of individual drugs among classes is crucial for precision medicine, which can allow clinicians to tailor treatment on the basis of patient groups. In this study, we did a comparative analysis of the 2 most frequently prescribed GLP-1 agonists, semaglutide (most frequent) and dulaglutide (second most frequent), against empagliflozin, a widely used SGLT-2 inhibitor (10). Using observational data from a large U.S. health care system with community and academic inpatient and outpatient practices, we structured the analysis as 2 hypothetical pragmatic randomized trials to emulate real-world treatment scenarios.

METHODS

Study Design

Using a retrospective observational cohort, we emulated 2 target trials of initial treatment with either semaglutide (primary trial) or dulaglutide (secondary trial) versus empagliflozin among adult patients with type 2 diabetes. The study followed the framework of target trial emulation, which aims to mimic the design and analysis of a hypothetical randomized controlled trial using observational data (11-13). The key study design and analysis components that were used to emulate the target trials are summarized in **Supplement Table 1** (available at Annals.org). The UPMC Quality Improvement Review Committee and Institutional Review Board provided ethical review and approval of the study as an exempt protocol, and all data remained deidentified for this analysis.

Data Sources

We used health-related data captured in the electronic health record (EHR) and ancillary clinical systems, aggregated and harmonized in a clinical data warehouse (14, 15). Specifically, patient information was linked by common identifiers across multiple EHRs, including MediPac, the admit, discharge, and transfer registration and hospital-based billing system; Cerner, the inpatient EHR for relevant clinical information for admitted patients at UPMC inpatient hospitals; and Epic, the UPMC EHR for ambulatory office visits owned by UPMC. For all patients, we accessed sociodemographic data, medical history information (such as comorbid conditions), laboratory results (such as HbA_{1c}), and anthropometric data (such as body mass index) from outpatient encounters. Use of medications was based on prescription orders made by physicians or other prescribers. Deaths were identified using hospital discharge dispositions of "ceased to breathe" sourced

from the inpatient medical record system; deaths after hospital discharge were identified via the Death Master File from the Social Security Administration's 2022 National Technical Information Service (16). Our health care system is exempt from the 3-year delay period by the Social Security Administration. Other nonfatal outcomes (such as MI) were obtained from inpatient encounters with diagnoses and procedures coded on the basis of the International Classification of Diseases, Ninth and Tenth Revisions, as previously published by our group (2, 17, 18). Because 97.8% of the source cohort self-declared their race as either Black (10.1%) or White (87.7%), we classified race as Black versus all others. In addition, only 1.5% of patients selfdeclared their ethnicity as Hispanic; hence, ethnicity was not used in the analyses.

Study Population (Eligibility Criteria)

All selected (eligible) patients had type 2 diabetes; were aged 45 years or older; and had EHR documentation of health care use, comorbid conditions, and recent use of prescription medications (for both diabetes and nondiabetes conditions) at the index prescription visit or at the closest office visit in the prior 5 years. Given very low prescription rates for semaglutide in our system in 2017 and 2018, the index date for patient selection was the first date between 1 January 2019 and 31 December 2024 in which a prescription for semaglutide, dulaglutide, or empagliflozin was documented in the EHR. As a potential washout (look-back) period, this selection was limited to patients without evidence of use of any other GLP-1 agonists or SGLT-2 inhibitors at the index office visit or in the prior 12 months (Supplement Figure 1, available at Annals.org). Concomitant use of other diabetes medications (for example, metformin or insulin) was permitted.

Exclusion criteria, which encompassed potential contraindications to semaglutide, dulaglutide, or empagliflozin (**Supplement Figure 1**), included type 1 diabetes, diabetic ketoacidosis, MI or stroke within the prior 6 months, thyroid cancer or family history of thyroid cancer, pancreatitis, multiple endocrine neoplasia syndrome type 2, current pregnancy, estimated glomerular filtration rate below 30 mL/min/1.73 m² or recent or current dialysis, and serious illness with an estimated 80% higher risk for death within 90 days (19). The study population was derived from 703 clinical practice groups located in southwestern Pennsylvania, northwestern Pennsylvania or New York, west central Pennsylvania or western Maryland, and north central Pennsylvania.

Treatment Strategies

Patients were classified as treated initially with either semaglutide or empagliflozin (primary trial) or dulaglutide or empagliflozin (secondary trial). Table 1. Comparison of Characteristics of Patients Treated With Semaglutide Versus Empagliflozin Before and After Matching*

Characteristic	stic Unmatched			Matched			
	Semaglutide (<i>n</i> = 14 068)	Empagliflozin (n = 13154)	SMD†	Semaglutide (n = 7899)	Empagliflozin (n = 7899)	SMD†	
Location of clinical practice	-	-	-	-	-	0.00*	
Altoona	1840 (13.1)	1233 (9.4)	0.12	867 (11.0)	867 (11.0)	-	
Center for Integrative Medicine	5569 (39.6)	5187 (39.4)	0.003	3144 (39.8)	3144 (39.8)	-	
Cole	179 (1.4)	232 (1.8)	0.04	88 (1.1)	88 (1.1)	-	
Erie Physicians Network	58 (0.4)	90 (0.7)	0.04	30 (0.4)	30 (0.4)	-	
Fayette Physician Network	14 (0.1)	76 (0.6)	0.08	10 (0.1)	10 (0.1)	-	
Great Lakes Physician Practice	763 (5.4)	340 (2.6)	0.14	276 (3.5)	276 (3.5)	-	
Renaissance Family Practice	309 (2.2)	374 (2.8)	0.04	139 (1.8)	139 (1.8)	-	
Soldiers and Sailors	28 (0.2)	17 (0.1)	0.02	6 (0.1)	6 (0.1)	-	
Somerset	2/8 (2.0)	462 (3.5)	0.09	210 (2.7)	210 (2.7)	-	
Susquehanna Health	863 (6.1)	800 (6.1)	0.002	531 (6.7)	531 (6.7)	-	
UPMC Hamot Regional Health Management System	1387 (9.9)	1476 (11.2)	0.04	860 (10.9)	860 (10.9)	-	
University of Pittsburgh Physicians	2081 (14.8)	2342 (17.8)	0.08	1419 (18.0)	1419 (18.0)	-	
Western Maryland	352 (2.5)	145 (1.1)	0.10	116 (1.5)	116 (1.5)	-	
All others	347 (2.5)	381 (2.9)	0.03	203 (2.6)	203 (2.6)	-	
Year started semaglutide or empagliflozin	-	-	-	-	-	0.00*	
treatment							
2019	//0 (5.5)	1604 (12.2)	0.24	645 (8.2)	645 (8.2)	-	
2020	853 (6.1)	1553 (11.8)	0.20	685 (8.7)	685 (8.7)	-	
2021	1911 (13.6)	2125 (16.1)	0.07	1331 (16.9)	1331 (16.9)	-	
2022	2532 (18.0)	2561 (19.5)	0.04	1536 (19.4)	1536 (19.4)	-	
2023	3690 (26.2)	2494 (19.0)	0.17	1728 (21.9)	1/28 (21.9)	-	
	4312 (30.6)	2817 (21.4)	0.21	1974 (25.0)	1974 (25.0)	-	
Mean age (SD) +	2407 (30.7)	5501 (42.3)	0.07	3213 (40.7)	3215 (40.7)	0.00"	
Mean age (SD), y	02.1 (7.3) 9024 (E7.0)	00.2 (10.3) E10E (20.9)	0.42	03.7 (7.0)	04.4 (7.7) 2425 (45.0)	0.07	
Plack race	1286 (0 1)	1222 (10.1)	0.37	786 (10.0)	202 (45.7) 202 (10 2)	0.00	
Commercial insurance	7709 (54.8)	6013 (45 7)	0.03	/019 (50.9)	3946 (50.0)	0.01	
Modicaro	6037 (42.9)	72/13 (55 1)	0.18	3769 (47.7)	3926 (19.7)	0.02	
Mean area deprivation index (SD)	68 2 (22 3)	66 9 (22 9)	0.25	677(223)	67 1 (22 9)	0.04	
Physician office visits in past year	00.2 (22.3)	024 (7.0)	0.00	G1.7 (22.3)	57((7.2)	0.004	
I or 2	737 (0.7) 2724 (24 4)	924 (7.0) 2525 (24.9)	0.01	2002 (24 4)	3/0(/.3) 207E(24/2)	0.004	
2.4	5/30 (20.0)	5286 (40.2)	0.01	2002 (20.4)	2073 (20.3)	0.001	
5-0 \7	2045 (28.2)	3/19 (26.0)	0.05	2208 (27.0)	2077 (26.2)	0.01	
ED visit in past year	457 (3.2)	506 (3.8)	0.03	272 (3.4)	274 (3 5)	0.02	
Mean body mass index (SD) ka/m^2	37 0 (7 2)	33 0 (6 5)	0.59	272 (3.4)	34 4 (6 6)	0.001	
Mean diastolic blood pressure (SD) mm Hg	78.2 (8.5)	76.8 (9.1)	0.17	77 8 (8 6)	77 5 (9 0)	0.14	
Mean hemoglobin A_1 (SD) %	7 2 (1 6)	76(15)	0.31	75(16)	7 5 (1 5)	0.05	
Medication use	,12 (110)	, 10 (110)	0.01	1.0 (1.0)	,10 (110)	0.00	
ACE inhibitor	5560 (39.5)	5854 (44.5)	0.10	3368 (42.6)	3397 (43.0)	0.01	
Angiotensin receptor-neprilysin inhibitor	77 (0.5)	569 (4.3)	0.25	75 (0.9)	65 (0.8)	0.01	
β -Blocker	4770 (33.9)	5854 (44.5)	0.20	2997 (37.9)	3050 (38.6)	0.01	
Insulin	3133 (22.3)	2749 (20.9)	0.03	1916 (24.3)	1879 (23.7)	0.01	
Metformin	9650 (68.6)	9534 (72.5)	0.09	5660 (71.7)	5692 (72.1)	0.01	
Statin	10847 (77.1)	11 192 (85.1)	0.21	6444 (81.6)	6583 (83.3)	0.05	
Sulfonylurea	2891 (20.6)	3851 (29.3)	0.20	2031 (25.7)	2131 (27.0)	0.03	
Charlson Comorbidity Index	1007 (7 2)	2487 (18 9)	0.35	833 (10.5)	857 (10 8)	0.01	
COPD	3923 (35.0)	3884 (29.5)	0.12	2622 (33.2)	2447 (31.0)	0.01	
Dementia	91 (0.6)	135 (1.0)	0.04	66 (0.8)	72 (0.9)	0.01	
Diabetes with comorbid conditions	4654 (33.1)	5874 (44.7)	0.24	3117 (39.5)	3215 (40.7)	0.03	
Elixhauser Comorbidity Index	,						
Depression	3481 (24.7)	2451 (18.6)	0.15	1743 (22.1)	1646 (20.8)	0.03	
Hypothyroidism	2970 (21.1)	2315 (17.6)	0.09	1564 (19.8)	1450 (18.4)	0.04	
Liver disease	2417 (17.2)	1841 (14.0)	0.09	1268 (16.1)	1231 (15.6)	0.01	
Lymphoma	68 (0.5)	121 (0.9)	0.05	51 (0.6)	43 (0.5)	0.01	
Peripheral vascular disease	1138 (8.1)	1482 (11.3)	0.11	738 (9.3)	750 (9.5)	0.01	
Rheumatoid arthritis	1178 (8.4)	811 (6.2)	0.09	582 (7.4)	538 (6.8)	0.02	
Valvular disease	986 (7.0)	1520 (11.6)	0.16	643 (8.1)	680 (8.6)	0.02	

Continued on following page

Table 1-Continued

Characteristic	Unmatched			Matched			
	Semaglutide (<i>n</i> = 14 068)	Empagliflozin (n = 13 154)	SMD†	Semaglutide (n = 7899)	Empagliflozin (n = 7899)	SMD†	
Mean Charlson Comorbidity Index total score (SD) Elixhauser AHRQ score	2.5 (1.6)	2.9 (1.8)	0.25	2.6 (1.7)	2.6 (1.7)	0.003	
Quartile 1	5081 (36.1)	3190 (24.3)	0.26	2451 (31.0)	2334 (29.6)	0.01	
Quartile 2	3148 (22.4)	2697 (20.5)	0.05	1758 (22.3)	1755 (22.2)	0.001	
Quartile 3	3410 (24.2)	3676 (27.9)	0.09	2082 (26.4)	2157 (27.3)	0.01	
Quartile 4	2429 (17.3)	3591 (37.3)	0.24	1608 (20.4)	1653 (20.9)	0.01	

ACE = angiotensin-converting enzyme; AHRQ = Agency for Healthcare Research and Quality; CHF = congestive heart failure; COPD = chronic obstructive pulmonary disease; ED = emergency department; SMD = standardized mean difference.

* Values are numbers (percentages) unless otherwise indicated. All variables were used in the propensity score model.

† Presented as absolute value.

Study Outcomes and Follow-up

The primary outcome was risk for the composite outcome of death, MI, or stroke over the longitudinal follow-up period that ended 31 December 2024. Secondary outcomes included the individual end points of all-cause death, MI, stroke, heart failure (HF), and atrial fibrillation, as well as the composite outcomes death or MI; MI or stroke; and any CVD event, defined as MI, stroke, HF, or atrial fibrillation. Tertiary outcomes included change in body weight and HbA_{1c} at 6 and 12 months. The follow-up period for the analysis started on the day after the index date (first day after the prescription for semaglutide, dulaglutide, or empagliflozin).

Statistical Analysis

We compared sociodemographic and clinical characteristics in patients treated with semaglutide or dulaglutide versus empagliflozin (before and after matching) using standardized mean differences (SMDs). We selected empagliflozin-treated patients matched to either semaglutide- or dulaglutide-treated patients using propensity score (PS) methods (20, 21). Specifically, we used PSs from a logistic regression model fitted with classification into the semaglutide or dulaglutide group as the response variable and inclusion of explanatory variables. Separate PS models were fitted for the primary analysis of semaglutide versus empagliflozin (Table 1) and secondary analysis of dulaglutide versus empagliflozin (Supplement Table 2, available at Annals.org). We used 1:1 PS greedy nearest-neighbor matching without replacement within a caliper width of 0.25 to construct the matched treated and nontreated groups. We matched patients directly on year of index treatment, clinical practice group, and membership or nonmembership in the UPMC health (insurance) plan. The latter was done for the theoretical (unknown) possibility of potential differences in provision of care, along with potential more thorough ascertainment of health status and clinical outcomes among UPMC health plan members (that is, patients most likely to use UPMC treatment facilities). We considered in the PS models the ZIP code-level area deprivation index, a validated neighborhood-level measure that is consistently associated with health outcomes (22).

From the matched groups, we used the Kaplan-Meier method to estimate incidence proportions (expressed as percentages) by initial treatment at 1, 2, and 3 years of follow-up. Incidence rates and incidence rate differences per 1000 person-years (with 95% Cls) were also calculated, along with corresponding hazard ratios (HRs) and 95% Cls. Because the propensity-matched treatment groups were generally well matched, no additional covariates were used for adjustment (that is, for potential residual confounding).

Missing values for key covariates with less than 15% missing at baseline were imputed using median values (**Supplement Table 3**, available at Annals.org). A recent HbA_{1c} value was missing in 28% of patients at baseline. Because this measure was viewed as critical for patient matching in the PS model by treatment received, the PS model included an indicator variable for missing, with remaining indicator variables of 4.0% to 5.6%, 5.7% to 6.4%, 6.5% to 8.0%, and greater than 8.0% (with missing coded as 0%). Assuming that HbA_{1c} values were missing at random, this approach seemed to be successful because HbA_{1c} values among matched patients with nonmissing HbA_{1c} values were nearly identical by treatment group (**Table 1**).

Exploratory subgroup analyses (that is, hypothesisgenerating) were done, including by age (45 to 64 vs. \geq 65 years); sex (female vs. male); treatment-naive (no vs. yes), defined as no use of the index medication before the 12-month look-back period; UPMC health plan member (no vs. yes); body mass index at baseline (<35 vs. ≥35 kg/m²); HbA_{1c} at baseline (<7% vs. ≥7%); insulin use at baseline (no vs. yes); metformin use at baseline (no vs. yes); and history of MI or cerebrovascular accident (no vs. yes). For all subgroup analyses, separate PS models were fitted to account for the specific characteristics most associated with imbalance by initial treatment received (Supplement Table 4, available at Annals.org). Cox proportional hazard models were fitted to formally test for interaction among subgroups. Supplement Table 5 (available at Annals.org) provides PS model results for all subgroups, as well as the primary and secondary target trial cohorts. A sensitivity analysis was done stratified by primary geographic location of the 703 clinical practice groups (Supplement Table 6, available at Annals.org).

Table 2. Primary and Secondary Outcomes in 1:1 PS-Matched Cohorts for Semaglutide Versus Empagliflozin and Dulaglutide Versus Empagliflozin

Outcome of Interest	Semaglutide vs. Empagliflozin		Dulaglutide vs. Empagliflozin		
	Semaglutide	Empagliflozin	Dulaglutide	Empagliflozin	
Death/MI/stroke (primary outcome)					
Events/patients at risk, n/N	398/7899	450/7899	494/6093	482/6093	
Median follow-up (IQR), y	2.20 (1.06 to 3.47)	2.17 (1.07 to 3.53)	3.02 (1.80 to 4.54)	3.04 (1.80 to 4.56)	
IR per 1000 PY	20.99	23.56	25.64	25.02	
Rate difference per 1000 PY (95% CI)	-2.56 (-5.56 to 0.44)	Reference	0.62 (-2.55 to 3.80)	Reference	
HR (95% CI)	0.89 (0.78 to 1.02)	Reference	1.03 (0.90 to 1.16)	Reference	
Death					
Events/patients at risk, n/N	260/7899	272/7899	324/6093	300/6093	
Median follow-up (IQR), y	2.24 (1.09 to 3.53)	2.22 (1.09 to 3.58)	3.10 (1.87 to 4.61)	3.12 (1.87 to 4.63)	
IR per 1000 PY	13.50	13.98	16.47	15.24	
Rate difference per 1000 PY (95% CI)	-0.49 (-2.82 to 1.85)	Reference	1.22 (-1.26 to 3.71)	Reference	
HR (95% CI)	0.97 (0.81 to 1.15)	Reference	1.08 (0.92 to 1.27)	Reference	
МІ					
Events/patients at risk, n/N	147/7899	175/7899	173/6093	171/6093	
Median follow-up (IQR), y	2.20 (1.07 to 3.48)	2.18 (1.07 to 3.55)	3.03 (1.82 to 4.56)	3.06 (1.82 to 4.58)	
IR per 1000 PY	7.73	9.12	8.93	8.83	
Rate difference per 1000 PY (95% CI)	-1.39 (-3.24 to 0.45)	Reference	0.10 (-1.77 to 1.98)	Reference	
HR (95% CI)	0.85 (0.68 to 1.05)	Reference	1.01 (0.82 to 1.25)	Reference	
Stroke					
Events/patients at risk n/N	16/7899	75/7899	81/6093	84/6093	
Modian follow.up (IOR) y	$222(1.08 \pm 0.351)$	$220(1.08 \pm 0.358)$	3 08 (1 84 to 4 60)	$3 10 (1.85 \pm 0.460)$	
IR per 1000 PY	2.22 (1.00 to 3.51)	3.87	A 1A	4 30	
Rate difference per 1000 PY (95% CI)	-1.48(-2.59 to -0.36)	Reference	-0.15(-1.44 to 1.13)	Reference	
HR (95% CI)	0.62 (0.43 to 0.89)	Reference	0.97 (0.71 to 1.31)	Reference	
Death/MI					
Events/patients at risk, <i>n/N</i>	368/7899	404/7899	446/6093	430/6093	
Median follow-up (IQR), y	2.20 (1.07 to 3.48)	2.18 (1.07 to 3.55)	3.03 (1.82 to 4.56)	3.06 (1.82 to 4.58)	
IR per 1000 PY	19.35	21.06	23.02	22.19	
Rate difference per 1000 PY (95% CI)	-1.71 (-4.56 to 1.14)	Reference	0.82 (-2.17 to 3.82)	Reference	
HR (95% CI)	0.92 (0.80 to 1.06)	Reference	1.04 (0.91 to 1.19)	Reference	
MI/stroke					
Events/patients at risk. n/N	186/7899	230/7899	243/6093	236/6093	
Median follow-up (IOR), v	2.20 (1.06 to 3.47)	2.17 (1.07 to 3.53)	3.02 (1.80 to 4.54)	3.04 (1.80 to 4.56)	
IR per 1000 PY	9.81	12.04	12.61	12.25	
Rate difference per 1000 PY (95% CI)	-2.23 (-4.33 to -0.13)	Reference	0.37 (-1.86 to 2.59)	Reference	
HR (95% CI)	0.81 (0.67 to 0.99)	Reference	1.03 (0.86 to 1.23)	Reference	
Heart failure					
Events/patients at risk n/N	307/7899	341/7899	319/6093	307/6093	
Median follow-up (IOR) v	2 16 (1 03 to 3 45)	2 13 (1 05 to 3 52)	2 99 (1 77 to 4 53)	3 03 (1 77 to 4 55)	
IR per 1000 PY	16.35	17.96	16.68	16.02	
Rate difference per 1000 PY (95% CI)	-1.61(-4.28 to 1.04)	Reference	1.65(-1.91 to 3.22)	Reference	
HR (95% CI)	0.91 (0.78 to 1.06)	Reference	1.04 (0.89 to 1.22)	Reference	
Atrial fibrillation					
Events/patients at risk n/N	71/7900	79/7900	85/6002	68/6003	
Median follow up (IOP) y	$222(1.09 \pm 0.251)$	$220(1.08 \pm 2.56)$	2 01 (1 84 +o 4 58)	3 10 (1 84 + 0.4 60)	
IR por 1000 PV	3 71	4.04	1 36	3.10 (1.04 t0 4.00)	
Rate difference per 1000 PV (95% CI)	0.22 (1.57 + 0.02)	4.04 Poforonco	(0.88) (0.36 to 2.12)	Deference	
HR (95% CI)	0.92 (0.66 to 1.26)	Reference	1.25 (0.91 to 1.72)	Reference	
A mu anulia a succes					
Any cardiac event	595/7900	667/7800	621/6002	642/6000	
Events/patients at risk, n//V	303/7077 2 40 (1 00 to 2 27)	00///077	031/0073	043/0073	
INIEGIAN TOHOW-UP (IQK), Y	2.0U(1.77 tO 3.37)	2.00 (1.00 to 3.43)	2.07 (1.00 to 4.41)	2.71 (1.00 to 4.42)	
Rate difference per 1000 PV (05% CI)	1.7/ 1.22 (0.00+~ 0.4E)	Deference	0.42(1.14)	S4.77 Poforonas	
HR (95% CI)	-4.22 (-0.00 to -0.45) 0.88 (0.79 to 0.99)	Reference	-0.03 (-4.41 to 3.10) 0.98 (0.88 to 1.10)	Reference	

HR = hazard ratio; IR = incidence rate; MI = myocardial infarction; PS = propensity score; PY = person-years.

We used SAS, version 9.4 (SAS Institute), for all analyses.

Role of the Funding Source

The American Heart Association had no role in the study's design, data collection, analysis, or manuscript preparation.

RESULTS

Baseline Characteristics

Before 1:1 PS matching, the mean age was 62.1 years among 14068 patients treated with semaglutide and 66.2 years among 13154 treated with empagliflozin (**Table 1**). From 2019 to 2024, use of semaglutide generally increased. Other substantial differences before matching (SMD \geq 0.30) for semaglutide-treated patients included higher prevalence of female sex (57.0% vs. 38.8%), higher mean body mass index (37.0 vs. 33.0 kg/m²), lower mean HbA_{1c} (7.2% vs. 7.6%), and lower prevalence of congestive HF (7.2% vs. 18.9%). The prevalence of comorbid conditions was generally lower in patients receiving semaglutide.

After 1:1 matching of 7899 patients treated with semaglutide and 7899 treated with empagliflozin, distributions of baseline characteristics were similar except for nominally higher body mass index in semaglutide-than empagliflozin-treated patients (35.2 vs. 34.4 kg/m²; SMD, 0.14). The median length of follow-up among matched patients who did not die was 2.3 years (IQR, 1.1 to 3.5 years) in semaglutide-treated patients versus 2.2 years (IQR, 1.1 to 3.6 years) in empagliflozin-treated patients. In the secondary trial analysis, after 1:1 PS matching of 6093 patients treated with dulaglutide and 6093 treated with empagliflozin, distributions of baseline characteristics were similar by treatment regimen (SMD <0.10 for all characteristics) (Supplement Table 2).

Primary Trial Results

Treatment History

For semaglutide-treated patients, rates of continued medication use based on available follow-up data at 6, 12, and 18 months were 75.1%, 64.6%, and 59.1%, respectively. Among these patients, corresponding rates of dulaglutide use were 5.3%, 7.4%, and 8.3% and corresponding rates of empagliflozin use were 3.8%, 6.5%, and 9.2%. For empagliflozin-treated patients, rates of continued medication use at 6, 12, and 18 months were 80.1%, 71.9%, and 67.4%, respectively. Among these patients, corresponding rates of semaglutide use were 4.8%, 8.8%, and 12.0% and corresponding rates of dulaglutide use were 2.2%, 4.2%, and 5.8%.

Effect of Semaglutide Versus Empagliflozin on Weight Loss and Glycemic Control

Mean changes in weight at 6 and 12 months were -3.6 kg (SD, 6.2) and -4.3 kg (SD, 7.6), respectively, for patients treated with semaglutide and -2.6 kg

(SD, 5.1) and -3.2 kg (SD, 6.2), respectively, for those treated with empagliflozin. Mean changes in HbA_{1c} at 6 and 12 months were -0.43 percentage point (SD, 1.37) and -0.34 percentage point (SD, 1.52), respectively, for patients treated with semaglutide and -0.21 percentage point (SD, 1.21) and -0.18 percentage point (SD, 1.32), respectively, for those treated with empagliflozin.

Effect of Semaglutide Versus Empagliflozin on MACE and Mortality

For the primary outcome, our results indicated a nominally lower cumulative incidence of death, MI, or stroke with the use of semaglutide versus empagliflozin starting at about 1 year of follow-up (Figure 1). Respective rates of death, MI, or stroke were 1.9% versus 2.2% at 1 year, 3.7% versus 4.5% at 2 years, and 5.9% versus 6.9% at 3 years. The incidence rate per 1000 person-years was 20.99 for semaglutide, compared with 23.56 for empagliflozin (Table 2, left). This resulted in a rate difference of -2.56 (95% Cl, -5.56 to 0.44) and HR of 0.89 (CI, 0.78 to 1.02). For the 7 secondary outcomes, all risk estimates were in the direction favoring use of semaglutide, with the lowest semaglutide-associated relative risks for stroke (HR, 0.62 [CI, 0.43 to 0.89]) and MI or stroke (HR, 0.81 [CI, 0.67 to 0.99]).

Exploratory Subgroup Analyses

Most subgroups examined with respect to risk for death, MI, or stroke yielded estimates in the direction that nominally favored treatment with semaglutide over empagliflozin (**Supplement Figure 2**, available at Annals.org). In the subgroups examined, the risk for death, MI, or stroke in relation to use of semaglutide was numerically lower among patients younger than 65 years (HR, 0.76 [CI, 0.61 to 0.96]) than among patients aged 65 years or older (HR, 1.02 [CI, 0.86 to 1.22]) (**Supplement Figure 1**). The risk for death, MI, or stroke in relation to use of semaglutide was also lower among patients with an HbA_{1c} below 7% (HR, 0.74 [CI, 0.56 to 0.97]) than among patients with an HbA_{1c} of 7% or higher (HR, 0.98 [CI, 0.81 to 1.19]).

Sensitivity Analysis

When the 703 clinical practice groups were categorized into 4 separate large geographic regions, the apparent lower risk for death, MI, or stroke associated with semaglutide was evident only among patients treated in clinical practices in southwestern Pennsylvania (HR, 0.75 [CI, 0.63 to 0.90]) (**Supplement Table 6**). In secondary analyses, patients from southwestern Pennsylvania (59.7% of the total sample), compared with all other locations, were more likely to self-report Black race (15.3% vs. 2.3%) and have higher indicators of access to health care, including higher enrollment in the UPMC health insurance plan (51.1% vs. 25.3%), higher use of telemedicine for office visits (15.4% vs. 3.5%), and higher annual median income (\$53 992 vs. \$45 644). *Figure 1.* Kaplan-Meier cumulative incidence curves of the risk for death/MI/stroke, by initial treatment with either semaglutide (*red*) or empagliflozin (*blue*).



MI = myocardial infarction.

Secondary Trial Results: Effect of Dulaglutide Versus Empagliflozin on Weight Loss, Glycemic Control, MACE, and Mortality

Mean changes in weight at 6 and 12 months were -2.0 kg (SD, 5.3) and -2.8 kg (SD, 6.7), respectively, for patients treated with dulaglutide and -2.4 kg (SD, 5.1) and -3.0 kg (SD, 6.2), respectively, for those treated with empagliflozin. Mean changes in HbA_{1c} at 6 and 12 months were -0.28 percentage point (SD, 1.40) and -0.25 percentage point (SD, 1.56), respectively, for patients treated with dulaglutide and -0.27 percentage point (SD, 1.31) and -0.27 percentage point (SD, 1.44), respectively, for those treated with empagliflozin.

The risk for death, MI, or stroke over follow-up was similar for patients treated with dulaglutide versus empagliflozin (**Figure 2**). Respective rates of death, MI, or stroke were 2.1% versus 2.4% at 1 year, 5.1% versus 4.5% at 2 years, and 8.1% versus 7.8% at 3 years. The incidence rate per 1000 person-years was 25.64 for dulaglutide-treated patients versus 25.02 for empagliflozin-treated patients (**Table 2**, *right*), with a corresponding HR of 1.03 (CI, 0.90 to 1.16) (**Table 2**, *right*). For the remaining secondary outcomes, the direction of cardiac risk estimates was mixed by treatment regimen, and none of the 95% CIs excluded the null value (0.0 for risk difference, 1.0 for HR).

DISCUSSION

In these head-to-head emulated target trials examining possible cardiovascular benefits of either semaglutide or dulaglutide versus empagliflozin among adult patients with type 2 diabetes aged 45 years and older from a large network of more than 700 clinical practice groups, we report 4 key findings. First, although subgroup estimates were imprecise, our results suggest that treatment with semaglutide (vs. empagliflozin) may be more beneficial for our primary composite outcome of death, MI, or stroke after an average of approximately 2 years of treatment, particularly among patients younger than 65 years. Second, this potential treatment advantage with semaglutide was largely driven by a clear lower risk for stroke (and potentially slight advantage over other cardiovascular outcomes) but not a clear lower risk for death. Third, treatment with dulaglutide (the second most prevalent GLP-1 agonist in this health care system) does not seem to confer any meaningful benefit over treatment with empagliflozin. Finally, the apparent modest treatment benefit of semaglutide over empagliflozin coincided with modest greater reductions in weight and HbA_{1c} within the first year of use. In aggregate, our findings are directly applicable to populations mostly of White race, with a low proportion of Hispanic ethnicity.

Consistent with a recent report by Lingvay and colleagues (23), semaglutide users in our study showed a greater weight loss and HbA_{1c} reduction than those receiving empagliflozin. Similarly, in the STEP-HFpEF (Semaglutide Treatment Effect in People with Obesity and HFpEF [HF with preserved ejection fraction]) trial, semaglutide improved HF-related symptoms, physical limitations, and exercise function, as well as levels of N-terminal pro-B-type natriuretic peptide, with the magnitude of benefit directly related to the extent of weight loss (24). It is thus plausible that the greater reduction in cardiovascular events (for example, MI or stroke) that we observed is explained at least in part by the weight loss and more robust glycemic control achieved by semaglutide users.

Although both GLP-1 agonists and SGLT-2 inhibitors have demonstrated efficacy in glycemic control, weight management, and cardiovascular benefits, they have different mechanisms of action and may confer distinct clinical advantages in different patient populations (25, 26). The latest clinical guidelines by the American College of Physicians (27) uphold such differences and, to date, offer several individualized recommendations for use of GLP-1 agonists and SGLT-2 inhibitors. However, more mechanistic and population-based data may be able to help delineate specific precision medicine approaches, especially with wider uptake and promising newer agents currently in advanced phases of clinical trials (28). In this regard, the current head-tohead analyses were done to potentially affect clinical decision making and optimize precision medicine initiatives.

At a molecular level, GLP-1 agonists stabilize endothelial function, reduce inflammation, and decrease oxidative stress (21, 22). However, we did not find superior effects in preventing the primary or HF outcome when comparing dulaglutide (second most commonly used GLP-1 in our health care system) versus empagliflozin. Although semaglutide and dulaglutide share the same primary mechanism of action with GLP-1 receptor agonists, several mechanisms may explain the possible superior cardiovascular profile of the former drug. Semaglutide has a consistently more pronounced effect on weight loss and glycemic control than dulaglutide in head-to-head comparisons (29-31). In addition, semaglutide has potent anti-inflammatory effects (32) and significantly reduced high-sensitivity C-reactive protein levels in patients with obesity (33), diabetes mellitus, or HF with preserved ejection fraction (34). Although no head-to-head comparisons are available between semaglutide and dulaglutide, recent data indicate that reductions in high-sensitivity C-reactive protein levels are more pronounced with semaglutide than with empagliflozin and other GLP-1 agonists (35).

Our suggestion of potential greater benefit in cardiovascular adverse outcome reductions among younger (<65 years) patients with semaglutide use warrants future investigation. One plausible explanation of this finding may be that there is more pronounced improvement in endothelial function, inflammation (36), and myocardial hemodynamics (37) in younger patients due to their generally better baseline cardiovascular health and greater capacity for physiologic adaptation. Further, younger patients generally have a better cardiovascular risk profile (38) and shorter exposure to these risk factors, which could influence a more meaningful semaglutide effect (39, 40). Younger patients may have better medication adherence due to more tolerance of the potential side effects of semaglutide than older patients. These reasons may explain the suggestion of greater benefit of semaglutide among younger patients in our study.

Finally, our data indicate a pronounced geographic influence on semaglutide outcomes among patients who had substantially greater access to health care,

Figure 2. Kaplan-Meier cumulative incidence curves of the risk for death/MI/stroke, by initial treatment with either dulaglutide (red) or empagliflozin (blue).



MI = myocardial infarction.

ORIGINAL RESEARCH

including higher insurance coverage, telemedicine use, and income. This suggests that regional differences likely stem from previously observed (41) disparities in health care access rather than a true geographic variation in treatment effect.

Our study has limitations. First, we could not assess cause-specific mortality, which would have provided additional insight into the relationship between diabetes treatment regimen and risk for CVD mortality. Second, classification of index medication use (semaglutide, empagliflozin, or dulaglutide) was based on prescription orders rather than prescriptions that were filled and did not account for the extent of adherence to filled prescriptions. Our suspicion is that any potential misclassification on medication use would tend to be nondifferential and bias results toward the null (no treatment association). Third, hospitalizations (such as for treatment of MI) that occurred outside the UPMC system were not captured in our analyses. However, because UPMC retains about 80% of its patient population, changes due to missing patient population may not be substantial. Furthermore, similar results were observed in the subgroup analysis of UPMC health plan members only (that is, the patients most likely to be treated in UPMC facilities). Finally, residual confounding may be present in our analysis.

In conclusion, among adult patients aged 45 years or older with type 2 diabetes, treatment with semaglutide seems to modestly lower the risk for our composite end point of death, MI, or stroke compared with treatment with empagliflozin; this finding was primarily driven by reductions in stroke and to a lesser degree MI risk. The advantage of semaglutide seems to be more evident among patients younger than 65 years and those with greater access to health care. Dulaglutide does not seem to confer a clinical treatment advantage over empagliflozin.

From School of Medicine, University of Pittsburgh, and Heart and Vascular Institute, UPMC, Pittsburgh, Pennsylvania (A.S., A.K., S.R., O.M.); School of Medicine, University of Pittsburgh; Heart and Vascular Institute, UPMC; and Clinical Analytics, UPMC, Pittsburgh, Pennsylvania (S.R.M.); Heart and Vascular Institute, UPMC, Pittsburgh, Pennsylvania (F.T.); and Clinical Analytics, UPMC, Pittsburgh, Pennsylvania (L.L., K.K.).

Grant Support: By American Heart Association grant 25GLP-1447886, used exclusively to cover publication fees.

Disclosures: Disclosure forms are available with the article online.

Reproducible Research Statement: *Study protocol:* No separate study protocol was required a priori because this retrospective analysis was deemed a quality improvement initiative with ethical review and approval granted by the UPMC Quality Improvement Review Committee and Institutional Review Board. A synopsis of

the study protocol that was submitted to and reviewed by the UPMC Quality Improvement Review Committee and Institutional Review Board may be made available on request to Dr. Saeed (e-mail, saeeda@pitt.edu). *Statistical code:* Relevant sections of the full statistical code used directly in the analysis (i.e., not for construction of the patient cohort) may be available to interested readers by contacting Dr. Kip (e-mail, kipke2@upmc.edu). *Data set:* The data set contains protected health information and will not be made available.

Corresponding Author: Anum Saeed, MD, UPMC Heart and Vascular Institute, University of Pittsburgh School of Medicine, 200 Lothrop Street, BST Building 1055W, Pittsburgh, PA 15213; e-mail, saeeda@pitt.edu.

Author contributions are available at Annals.org.

References

1. Martin SS, Aday AW, Almarzooq ZI, et al; American Heart Association Council on Epidemiology and Prevention Statistics Committee and Stroke Statistics Subcommittee. 2024 heart disease and stroke statistics: a report of US and global data from the American Heart Association. Circulation. 2024;149:e347-e913. [PMID: 38264914] doi:10.1161/CIR.00000000001209

2. Muluk P, Zhu J, Thoma F, et al. Impact of guideline-directed statin intervention for primary prevention in patients with diabetes. Diabetes Care. 2023;46:2273-2277. [PMID: 37851356] doi:10.2337/dc23-0816

3. Mohseni-Moghaddam P, Ghobadian R, Khaleghzadeh-Ahangar H. Dementia in diabetes mellitus and atherosclerosis: two interrelated systemic diseases. Brain Res Bull. 2022;181:87-96. [PMID: 35093470] doi:10.1016/j.brainresbull.2022.01.018

4. Packer M, Anker SD, Butler J, et al. Effect of empagliflozin on the clinical stability of patients with heart failure and a reduced ejection fraction: the EMPEROR-Reduced trial. Circulation. 2021;143:326-336. [PMID: 33081531] doi:10.1161/CIRCULATIONAHA.120.051783 5. Zinman B, Wanner C, Lachin JM, et al; EMPA-REG OUTCOME Investigators. Empagliflozin, cardiovascular outcomes, and mortality in type 2 diabetes. N Engl J Med. 2015;373:2117-2128. [PMID: 26378978] doi:10.1056/NEJMoa1504720

6. Neal B, Perkovic V, Mahaffey KW, et al; CANVAS Program Collaborative Group. Canagliflozin and cardiovascular and renal events in type 2 diabetes. N Engl J Med. 2017;377:644-657. [PMID: 28605608] doi:10.1056/NEJMoa1611925

7. Das SR, Everett BM, Birtcher KK, et al. 2020 expert consensus decision pathway on novel therapies for cardiovascular risk reduction in patients with type 2 diabetes: a report of the American College of Cardiology Solution Set Oversight Committee. J Am Coll Cardiol. 2020;76:1117-1145. [PMID: 32771263] doi:10.1016/ j.jacc.2020.05.037

8. American Diabetes Association Professional Practice Committee. 10. Cardiovascular disease and risk management: Standards of Medical Care in Diabetes–2022. Diabetes Care. 2021;45:S144-S174. [PMID: 34964815] doi:10.2337/dc22-S010

9. Drake T, Landsteiner A, Langsetmo L, et al. Newer pharmacologic treatments in adults with type 2 diabetes: a systematic review and network meta-analysis for the American College of Physicians. Ann Intern Med. 2024;177:618-632. [PMID: 38639549] doi:10.7326/ M23-1490

10. American Diabetes Association Professional Practice Committee. 8. Obesity and weight management for the prevention and treatment of type 2 diabetes: Standards of Care in Diabetes–2024. Diabetes Care. 2024;47:S145-S157. [PMID: 38078578] doi:10.2337/dc24-S008

Original Research

11. Hansford HJ, Cashin AG, Jones MD, et al. Reporting of observational studies explicitly aiming to emulate randomized trials: a systematic review. JAMA Netw Open. 2023;6:e2336023-e2336023. [PMID: 37755828] doi:10.1001/jamanetworkopen.2023.36023

12. Hernán MA, Robins JM. Using big data to emulate a target trial when a randomized trial is not available. Am J Epidemiol. 2016;183: 758-764. [PMID: 26994063] doi:10.1093/aje/kwv254

13. Hernán MA, Sauer BC, Hernández-Díaz S, et al. Specifying a target trial prevents immortal time bias and other self-inflicted injuries in observational analyses. J Clin Epidemiol. 2016;79:70-75. [PMID: 27237061] doi:10.1016/j.jclinepi.2016.04.014

14. Reitz KM, Seymour CW, Vates J, et al. Strategies to Promote ResiliencY (SPRY): a randomised embedded multifactorial adaptative platform (REMAP) clinical trial protocol to study interventions to improve recovery after surgery in high-risk patients. BMJ Open. 2020;10:e037690. [PMID: 32994242] doi:10.1136/bmjopen-2020-037690

15. Bariola JR, McCreary EK, Wadas RJ, et al. Impact of bamlanivimab monoclonal antibody treatment on hospitalization and mortality among nonhospitalized adults with severe acute respiratory syndrome coronavirus 2 infection. Open Forum Infect Dis. 2021;8: ofab254. [PMID: 34250192] doi:10.1093/ofid/ofab254

16. **Social Security Administration**. Requesting SSA's death information. Accessed at www.ssa.gov/dataexchange/request_dmf on 16 September 2024.

17. Walker AJ, Zhu J, Thoma F, et al. Statin utilization and cardiovascular outcomes in a real-world primary prevention cohort of older adults. Am J Prev Cardiol. 2024;18:100664. [PMID: 38665251] doi:10.1016/j. ajpc.2024.100664

18. Saeed A, Zhu J, Thoma F, et al. Cardiovascular disease riskbased statin utilization and associated outcomes in a primary prevention cohort: insights from a large health care network. Circ Cardiovasc Qual Outcomes. 2021;14:Circoutcomes120007485. [PMID: 34455825] doi:10.1161/CIRCOUTCOMES.120.007485

19. **Oo TH, Marroquin OC, McKibben J, et al.** Improved palliative care practices through machine-learning prediction of 90-day risk of mortality following hospitalization. NEJM Catal Innov Care Deliv. 2022;4. [CAT.22.0214]

20. Austin PC. An introduction to propensity score methods for reducing the effects of confounding in observational studies. Multivariate Behav Res. 2011;46:399-424. [PMID: 21818162] doi:10.1080/00273171.2011.568786

21. Rosenbaum PR, Rubin DB. The central role of the propensity score in observational studies for causal effects. Biometrika. 1983;70:41-55. doi:10.1093/biomet/70.1.41

22. Kind AJH, Buckingham WR. Making neighborhood-disadvantage metrics accessible – the Neighborhood Atlas. N Engl J Med. 2018; 378:2456-2458. [PMID: 29949490] doi:10.1056/NEJMp1802313

23. Lingvay I, Capehom MS, Catarig A-M, et al. Efficacy of once-weekly semaglutide vs empagliflozin added to metformin in type 2 diabetes: patient-level meta-analysis. J Clin Endocrinol Metab. 2020;105: e4593-e4604. [PMID: 32827435] doi:10.1210/clinem/dgaa577

24. Petrie MC, Borlaug BA, Butler J, et al; STEP-HFpEF Trial Committees and Investigators. Semaglutide and NT-proBNP in obesity-related HFpEF: insights from the STEP-HFpEF program. J Am Coll Cardiol. 2024;84:27-40. [PMID: 38819334] doi:10.1016/j. jacc.2024.04.022

25. Gurgle HE, White K, McAdam-Marx C. SGLT2 inhibitors or GLP-1 receptor agonists as second-line therapy in type 2 diabetes: patient selection and perspectives. Vasc Health Risk Manag. 2016;12:239-249. [PMID: 27350752] doi:10.2147/VHRM.S83088

26. Buse JB, Wexler DJ, Tsapas A, et al. 2019 update to: management of hyperglycemia in type 2 diabetes, 2018. a consensus report by the American Diabetes Association (ADA) and the European Association for the Study of Diabetes (EASD). Diabetes Care. 2020;43:487-493. [PMID: 31857443] doi:10.2337/dci19-0066 27. Qaseem A, Obley AJ, Shamliyan T, et al; Clinical Guidelines Committee of the American College of Physicians. Newer pharmacologic treatments in adults with type 2 diabetes: a clinical guideline from the American College of Physicians. Ann Intern Med. 2024;177:658-666. [PMID: 38639546] doi:10.7326/M23-2788

28. Jastreboff AM, Kaplan LM, Frías JP, et al; Retatrutide Phase 2 Obesity Trial Investigators. Triple-hormone-receptor agonist retatrutide for obesity – a phase 2 trial. N Engl J Med. 2023;389:514-526. [PMID: 37366315] doi:10.1056/NEJMoa2301972

29. Pratley RE, Aroda VR, Catarig A-M, et al. Impact of patient characteristics on efficacy and safety of once-weekly semaglutide versus dulaglutide: SUSTAIN 7 post hoc analyses. BMJ Open. 2020;10: e037883. [PMID: 33199417] doi:10.1136/bmjopen-2020-037883

30. Pratley RE, Aroda VR, Lingvay I, et al; SUSTAIN 7 investigators. Semaglutide versus dulaglutide once weekly in patients with type 2 diabetes (SUSTAIN 7): a randomised, open-label, phase 3b trial. Lancet Diabetes Endocrinol. 2018;6:275-286. [PMID: 29397376] doi:10.1016/S2213-8587(18)30024-X

31. Pratley RE, Catarig A-M, Lingvay I, et al. An indirect treatment comparison of the efficacy of semaglutide 1.0 mg versus dulaglutide 3.0 and 4.5 mg. Diabetes Obes Metab. 2021;23:2513-2520. [PMID: 34286894] doi:10.1111/dom.14497

32. Masson W, Lobo M, Nogueira JP, et al. Anti-inflammatory effect of semaglutide: updated systematic review and meta-analysis. Front Cardiovasc Med. 2024;11:1379189. [PMID: 39055657] doi:10.3389/ fcvm.2024.1379189

33. Newsome P, Francque S, Harrison S, et al. Effect of semaglutide on liver enzymes and markers of inflammation in subjects with type 2 diabetes and/or obesity. Aliment Pharmacol Ther. 2019;50:193-203. [PMID: 31246368] doi:10.1111/apt.15316

34. Kosiborod MN, Abildstrøm SZ, Borlaug BA, et al; STEP-HFpEF Trial Committees and Investigators. Semaglutide in patients with heart failure with preserved ejection fraction and obesity. N Engl J Med. 2023;389:1069-1084. [PMID: 37622681] doi:10.1056/NEJMoa2306963

35. Mosenzon O, Capehorn MS, De Remigis A, et al. Impact of semaglutide on high-sensitivity C-reactive protein: exploratory patient-level analyses of SUSTAIN and PIONEER randomized clinical trials. Cardiovasc Diabetol. 2022;21:172. [PMID: 36056351] doi:10.1186/ s12933-022-01585-7

36. Lin K, Wang A, Zhai C, et al. Semaglutide protects against diabetes-associated cardiac inflammation via Sirt3-dependent RKIP pathway. Br J Pharmacol. 2025;182:1561-1581. [PMID: 39710830] doi:10.1111/bph.17327

37. Stone C, Harris DD, Broadwin M, et al. Semaglutide improves myocardial perfusion and performance in a large animal model of coronary artery disease. Arterioscler Thromb Vasc Biol. 2025;45:285-297. [PMID: 39665144] doi:10.1161/ATVBAHA.124.321850

38. Saeed A, Nambi V, Sun W, et al. Short-term global cardiovascular disease risk prediction in older adults. J Am Coll Cardiol. 2018;71:2527-2536. [PMID: 29535064] doi:10.1016/j.jacc.2018.02.050

39. Deanfield J, Verma S, Scirica BM, et al; SELECT Trial Investigators. Semaglutide and cardiovascular outcomes in patients with obesity and prevalent heart failure: a prespecified analysis of the SELECT trial. Lancet. 2024;404:773-786. [PMID: 39181597] doi:10.1016/S0140-6736(24)01498-3

40. Nauck MA, Quast DR. Cardiovascular safety and benefits of semaglutide in patients with type 2 diabetes: findings from SUSTAIN 6 and PIONEER 6. Front Endocrinol (Lausanne). 2021;12:645566. [PMID: 33854484] doi:10.3389/fendo.2021.645566

41. Lu Y, Liu Y, Krumholz HM. Racial and ethnic disparities in financial barriers among overweight and obese adults eligible for semaglutide in the United States. J Am Heart Assoc. 2022;11:e025545. [PMID: 36172953] doi:10.1161/JAHA.121.025545 **Author Contributions:** Conception and design: A. Saeed, K. Kip. Analysis and interpretation of the data: S.R. Mulukutla, L. Lemon, K. Kip.

Drafting of the article: A. Saeed, L. Lemon, A. Koczo, S. Reis, K. Kip.

Critical revision for important intellectual content: A. Saeed, S.R. Mulukutla, L. Lemon, O. Marroquin, K. Kip.

Final approval of the article: A. Saeed, S.R. Mulukutla, F. Thoma, L. Lemon, A. Koczo, S. Reis, O. Marroquin, K. Kip.

Statistical expertise: L. Lemon, K. Kip.

Collection and assembly of data: S.R. Mulukutla, F. Thoma, K. Kip.